Author's response to reviews

Title: Gastroprotective effect of carob (Ceratonia siliqua L.) against ethanol-induced oxidative stress in rat

Authors:

- Kais Rtibi (rtibikais@yahoo.fr)
- Abdelaziz Souli (AbdelazizSouli@fsg.rnu.tn)
- Hichem Sebai (sebai.Hichem@yahoo.fr)
- Mohamed Amine Jabri (jabriamino@gmail.com)
- Slimen Selmi (slimen.selmi@gmail.com)
- Jamel El-Benna (jamelelbenna@inserm.fr)
- Mohamed Amri (mohamedamri@fst.rnu.tn)
- Mohamed Lamjid Marzouki (lamjed.marzouki@ipeis.rnu.tn)

Version: 6 Date: 6 May 2015

Author's response to reviews: see over
Dear Editor

Thank you for your positive response about our manuscript (MS: 8625288311597487). I would like to thank again the editor and all the reviewers who contributed to revise our manuscript to become worth for publication. Please note that all the recommendations suggested by the reviewers have been followed point by point in the revised manuscript. All modifications added in the revised manuscript were highlighted in Blue colour.

Enclose with this cover letter an explanatory report and a revised manuscript.

Hope to get from you a positive response.

Best regards,

Sincerely,

Dr. Kais Rtibi
Laboratoire de Neurophysiologie Fonctionnelle et Pathologies, Département des Sciences Biologiques, Faculté des Sciences de Tunis, Campus Universitaire El Manar II-2092 Tunis, Tunisia.
E-mail: rtibikais@yahoo.fr
Fax: +(216) 72 590 566
Tel: +(216) 97 479 135